Author's response to reviews

Title: Mixed Messages or Lack of Consensus? A qualitative study of the impact of the Prescribing Analysis System on the prescribing behaviors of physicians in Korea

Authors:

Dong-Sook Kim (sttone@hiramail.net)
Green Bae (greeni77@gmail.com)
Soo Yeon Yoo (yousooyeon@hiramail.net)
Minah Kang (minahkang@ewha.ac.kr)

Version: 2 Date: 9 December 2014

Author's response to reviews: see over
December 9, 2014

Dear Editor-in-Chief,

Thank you very much for your important comments on our manuscript. We are very pleased to have this opportunity to refine our article. We added line and page numbers and a declaration of competing interests. Please find our revised manuscript, "Mixed Messages or Lack of Consensus? A qualitative study of the impact of the Prescribing Analysis System (PAS) on the prescribing behaviors of physicians in Korea."

Thank you in advance for your kind consideration. We look forward to your reply. Please address all correspondence to minahkang@ewha.ac.kr.

We look forward to hearing from you at your earliest convenience.

Sincerely yours,

Minah Kang, PhD
Associate Professor
Department of Administration,
College of Social Sciences,
Ewha Womans University
Room 314, 52, Ewhayeodae-gil, Seodaemun-gu, Seoul 120-750,
South Korea
Tel: +82-2-3277-4100
E-mail: minahkang@ewha.ac.kr